Author's response to reviews

Title: The Tibetan herbal medicines Padma 28 and Padma Circosan inhibit the formation of advanced glycation endproducts (AGE) and advanced oxidation protein products (AOPP) in vitro.

Authors:

Ewa Grzebyk (ewa.grzebyk@gmail.com)
Agnieszka Piwowar (agnieszka.piwowar@umed.wroc.pl)

Version: 3
Date: 21 July 2014

Author's response to reviews: see over
Title: The Tibetan herbal medicines Padma 28 and Padma Circosan inhibit the formation of advanced glycation endproducts (AGE) and advanced oxidation protein products (AOPP) in vitro.
MS: 1471078235127953
Authors: Ewa Grzebyk, Agnieszka Piwowar

Reply to editors comment

Date: 21 July 2014

Dear Editors of BMC Complementary & Alternative Medicine

I apologize for such a long time response to a letter from you, I was on holidays and as soon as I got back to work I respond to the letter.

In response to your comments on Competing Interests section: we have included an appropriate information in the article (page 9 lines 8-10).
The manufacturer of Padma 28 and Padma Circosan provided the test materials. The manufacturer was not involved in the study design, the analytical performance, the analysis of the data or the preparation of the manuscript. The study was financed from own resources of the Wroclaw Medical University (Pbmn175), and Padmas manufacturers was co-financing only the purchase of immunoenzymatic tests (ELISA tests).

Your sincerely
Ewa Grzebyk and Agnieszka Piwowar.